Author's response to reviews

Title: The use of advanced tracking technologies for the analysis of mobility in Alzheimer's disease and related cognitive diseases

Authors:

Noam Shoval (noamshoval@huji.ac.il)
Gail K Auslander (msgail@pluto.mscc.huji.ac.il)
Tim Freytag (Tim.Freytag@urz.uni-heidelberg.de)
Ruth Landau (mslanda@mscc.huji.ac.il)
Frank Oswald (oswald@dzfa.uni-heidelberg.de)
Ulrich Seidl (Ulrich.seidl@med.uni-heidelberg.de)
Hans-Werner Wahl (wahl@dzfa.uni-heidelberg.de)
Shirli Werner (shirliw@bezeqint.net)
Jeremia Heinik (heinik@post.tau.ac.il)

Version: 2 Date: 27 February 2008

Author's response to reviews:

Enclosed please find our revised manuscript for review. In this revised version we have included a power calculation for sample size. Further, we have ensured that the revised manuscript conforms to the journal style.

We hope that in its present form you will find this manuscript suitable for publication in BMC Geriatrics.